Author’s response to reviews

Title: Unsuspected Pneumocystis pneumonia in an HIV-seronegative patient with untreated lung cancer: a case report

Authors:
Cai Chuang (skinblack1966@yahoo.com.cn)
Xie Zhanhong (Drxiezhanhong@163.com)
Gu Yinyin (DrGuinyin@163.com)
Zhong Shuqing (Drzhongshuqing@163.com)
Zhong Nanshan (DrNanshanzhong@163.com)

Version: 7 Date: 18 October 2007

Author’s response to reviews:

Dear Editor,

We have made a few minor modifications to the last manuscript, in accordance with your instructions. First, the initials of the authors like M.D and Ph.D are added back to each author along with the lastes email addresses of Drs Gu yinyin and Zhong nanshan, we are sorry to find, Dr Zeng qingsi, the radiologist, had been left out in the author list, of course, now his name is included in the author list. Secondly, we removed the line numbers as instructed, and corrected one more spelling mistake “turmour” back to “tumour” in the text. Thirdly, declaration on patient consent was added in the acknowledge section as instructed.Lastly, the "et al"s in the reference section was replaced by the full account of the authors, as instructed.

Best wishes.
Caichuang 2007-10-18